Author's response to reviews

Title: Acute illness associated with Ehrlichia sp. P-Mtn from Atlanta, Georgia, USA: a case report

Authors:

Will K Reeves (will.reeves@ars.usda.gov)
Amanda D Loftis (adloftis@gmail.com)
William L Nicholson (wln4@cdc.gov)
Alan G Czarkowski (bkt3@cdc.gov)

Version: 2 Date: 22 January 2008

Author's response to reviews:

Revised to conform fully to JMCR journal format. The authors apologize for the confusion; the instructions for authors on the JMCR web site (http://jmedicalcasereports.com/manuscript/case/info) were not as detailed as the manuscript template document provided by editorial staff, especially in regards to title page and abstract formatting.

We hope this will be more appropriate.

Thank you,
AD Loftis